Author’s response to reviews

Title: Traumatic events: Exploring associations with maternal depression, infant bonding, and oxytocin in Latina mothers

Authors:

Sandraluz Lara-Cinisomo (laracini@illinois.edu)
Kefu Zhu (kfz0131@gmail.com)
Kexin Fei (513331855@qq.com)
Yumeng Bu (yumengbu@gmail.com)
Alexandria Weston (a.weston17@yahoo.com)
Uma Ravat (uma.ravat.uiuc@gmail.com)

Version: 2 Date: 30 Aug 2017

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to revise our submission. Below is our response to each request:

1. Please provide separate Discussion and Conclusion sections in your main manuscript. You would NOT have to rewrite your manuscript, but please just insert Discussion and Conclusion headings at appropriate places in your manuscript.

Response: We have added the requested headers.

2. Please revise your Authors’ contribution section to confirm all authors have read and approved of the final manuscript.

Response: We have added the requested statement.